Author's response to reviews

Title: Household Exposure to Violence and Human Rights Violations in Western Bangladesh (II): History of Torture and Other Traumatic Experience of Violence and Functional Assessment of Victims

Authors:

Shr-Jie Wang (sjw@rcf.dk)
Mohammad Akramul Haque (akramul2000@yahoo.com)
Saber-ud-Daula Masum (liton.masum@yahoo.com)
Shuvodwip Biswas (shuvo85@yahoo.com)
Jens Modvig (jmo@rcf.dk)

Version: 6 Date: 2 November 2009

Author’s response to reviews:

We have made all major and minor changes suggested by the editors. Grammar mistakes and typing errors were checked. The blank page 38 was removed. We confirm that all authors have read and approved the final version of the manuscript. We had used sentence case in the title and format the abbreviations as a paragraph. All the tables fit into A4 pages the table title is only one sentence long. Additional information was treated as a legend and it appear below the table. We have also removed the visible vertical lines from all the tables. In the original figure 1a, there were gaps in years on the x-axis and we have made the corrections, then it showed all years from 1971-Feb. 2009.